Author's response to reviews

Title: What is behind smoker support for new smokefree areas? National survey data

Authors:

Nick Wilson (nWilson@actrix.gen.nz)
Deepa Weerasekera (deepa.w@xtra.co.nz)
Tony Blakely (tony.blakely@otago.ac.nz)
Richard Edwards (richard.edwards@otago.ac.nz)
George Thomson (george.thomson@otago.ac.nz)
Heather Gifford (heather.whakauae@xtra.co.nz)

Version: 4 Date: 17 August 2010

Author's response to reviews:

Dear Editors

1. Reference 49: please provide a functioning link to this website.

Response: This URL has been updated.

2. Tables: the first table referenced in the manuscript text should be labelled Table 1. The second, Table 2, etc. The tables should be provided in this same order at the end of the manuscript file. Please do not use ranges when listing tables. Currently the tables are cited out of numerical order.

Response: Changes made.

3. Authors’ contributions: Please use the following format as an example, including using initials to refer to each author's contribution: AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript.

Response: Changes made.

4. Authors’ contributions: Please indicate that all authors read and approved the final manuscript.

Response: Done (in authors’ contributions)

5. Affiliations: Please remember to include the country in the affiliation details.

Response: Done
6. Competing interests: Please insert the competing interests before the acknowledgements section.

Response: Done

7. Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures.

Response: Done

8. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Response: One minor typo corrected and some reference formatting improved.

Thank you for the opportunity to amend this manuscript.

Regards

Nick Wilson